Computer Science 505

Autumn 1994

Course Title: Concepts of Programming Languages

Instructor: Alan Borning

Sieg 426C 543-6678

b orning@cs.washington.edu

O�ce Hours: Monday 4:30{5:20, W ednesday 2:30{3:20, or whenever the do or is op en

T A: Ben Dugan

brd@cs.washington.edu

O�ce Hours: to b e arranged

T ext: none | we’ll use pap ers from the literature

Class Meetings: MWF 11:30 - 12:20, EEB 108

The required work for the course will b e homework (more or less weekly), a midterm, a �nal, and a course

pro ject. The homework will include some small programming pro jects in a numb er of languages, including

Ada, Prolog, Miranda, Smalltalk-80, CLP(R), Strand, and Oz, as well as some problem sets and essay topics.

Smalltalk-80 is on the MSCC Macintoshes; the others are on the department’s instructional machines lynx,

wolf, and grizzly

. The pro ject will involve writing a 4{10 page pap er on a topic related to programming

language design and implementation, and may also involve a programming pro ject. Students can work

individually or in pairs on the pro ject and pap er. Optionally

, students can give a presentation in class on

their work. (More work will b e exp ecte d of a pair than an individual.)

cse505@cs.washington.edu is a mailing list for the class|miscellaneous announcements will b e sent out on

the list, and you can use it yourselves to send comments or questions to the entire class. Send a message to

Ben for changes to the list.

I will usually give out printed copies of lecture notes in adv ance. W e’re also going to get these set up on

Mosaic; details shortly .

T entatively

, the grading for 505 will b e:

15% midterm

25% �nal

35% homework

25% pro ject